Author's response to reviews

Title: An intronic alteration of the fibroblast growth factor 10 gene causing ALSG-(aplasia of lacrimal and salivary glands) syndrome

Authors:

Kathrin Scheckenbach (scheckenbach@med.uni-duesseldorf.de)
Vera Balz (balz@med.uni-duesseldorf.de)
Martin Wagenmann (martin.wagenmann@uni-duesseldorf.de)
Thomas K Hoffmann (thomas.hoffmann@med.uni-duesseldorf.de)

Version: 12 Date: 14 December 2008

Author's response to reviews:

Dear editors,
small typographical errors were corrected.